Author's response to reviews

Title: Randomised controlled trial of a psychiatric consultation model for treatment of common mental disorder in the occupational health setting. Study Protocol.

Authors:

Christina M Van der Feltz-Cornelis (cfeltz@trimbos.nl)
Fransina J De Jong (sjong@trimbos.nl)
Rob Hoedeman (rob.hoedeman@arboned.nl)
Jolanda Meeuwissen (jmeeuwissen@trimbos.nl)
Iman Elfeddali (ielfeddali@trimbos.nl)

Version: 3  Date: 22 December 2006

Author's response to reviews: see over
Concerns: letters of approval of fund and medical ethical committee.

Dear Editor,

As you asked, I provide now the letters of approval of the fund for the grant application and of the medical ethical committee. On the first page, you can read:

"De Adviesraad is tot een positief oordeel gekomen". This means that the advisory board of the fund has given a positive approval to the grant application for the trial.

On the fourth page, you can read:
"De commissie besluit positief over uitvoering van dit onderzoek". This means that the Medical ethical committee decides positive concerning conducting this trial.

I do hope that this way, you are sufficiently informed to decide to publish the article as a design study without further review process.

If you have any more questions, please let us know.

Sincerely,

Dr. Christina M. van der Feltz-Cornelis
Corresponding author